

Author's response to reviews

Title: Statistical estimates of absenteeism attributable to seasonal and pandemic influenza from the Canadian Labour Force Survey

Authors:

Dena L Schanzer (dena.schanzer@phac-aspc.gc.ca)
Hui Zheng (hui.zheng@phac-aspc.gc.ca)
Jason Gilmore (Jason.Gilmore@statcan.gc.ca)

Version: 5 **Date:** 25 March 2011

Author's response to reviews:

Ref: MS: 1233077791451727

Statistical estimates of absenteeism attributable to seasonal and pandemic influenza from the Canadian Labour Force Survey

Dena L Schanzer, Hui Zheng and Jason Gilmore

25 March, 2011

Dear BMC Series Editorial Production Team, BioMed Central

As requested, the title for Figure 4 was revised.

Yours sincerely,

Dena Schanzer
Corresponding author
Public Health Agency of Canada